Author's response to reviews

Title: Factors associated with 2009 pandemic influenza A (H1N1) vaccination acceptance among university students from India during the post-pandemic phase

Authors:

Padmanaban S Suresh (suresh.ps@vit.ac.in)
Venkatesh Thejaswini (thejaswini_v@mrdg.iisc.ernet.in)
Thangaraju Rajan (rajanavmc@gmail.com)

Version: 3 Date: 28 July 2011

Author's response to reviews:

Dear Editor:

Thanks for giving me an opportunity to submit our article. We have made necessary changes in the article as per the suggestion.

Looking forward in hearing from you.

Please find the responses below

Major revisions (we require the author to make these changes)

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

We have made the tables as per the advice.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

We have removed all the highlighting in the revised version.

References: the reference list should contain up to 30 authors' names for each
citation. The term 'et al.' should not be used if there are fewer than 30 authors.

We have checked and corrected it

Tables- please remove the visible vertical lines from your tables

We have removed the vertical lines

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked for typographical errors to the best of our limits.

sincerely

suresh.